Author's response to reviews

Title: "If the patients decide not to tell what can we do?" - TB/HIV counsellors' dilemma on partner notification for HIV

Authors:

Barnabas N Njozing (njozing.barnabas@epiph.umu.se)
Kerstin E Edin (kerstin.edin@epiph.umu.se)
Miguel San Sebastián (miguel.sansebastian@epiph.umu.se)
Anna-Karin Hurtig (anna-karin.hurtig@epiph.umu.se)

Version: 3 Date: 24 May 2011

Author's response to reviews: see over
Point-by-point description of formatting changes

**Major revisions**

1. On page 25, paragraph 2; the ‘conclusion and policy implications’ section has been changed to ‘conclusions’.
2. All the highlighting and tracking in the manuscript have been removed.
3. A functional link to reference 35 on page 31 has been provided.

**Minor revisions**

1. The title and figure number of ‘Figure 1’ have been removed. The title of the Figure 1 has been included on page 34 of the manuscript after the reference list.
2. Both the table and figure have been formatted to match the journal recommendation.
3. The entire manuscript has been properly proof-read while checking for typographical and grammatical errors.